Author’s response to reviews

Title: Improving the prediction of the trabecular bone microarchitectural parameters using dental cone-beam computed tomography

Authors:

Rong-Ting He (carnea101@gmail.com)
Ming-Gene Tu (mgtu@mail.cmu.edu.tw)
Heng-Li Huang (hlhuang@mail.cmu.edu.tw)
Ming-Tzu Tsai (anniemtt@sunrise.hk.edu.tw)
Jay Wu (iamjaywu@gmail.com)
Jui-Ting Hsu (jthsu@mail.cmu.edu.tw;richard@ms32.url.com.tw)

Version: 4 Date: 14 Jan 2019

Author’s response to reviews:

Senior Assistant Editor Comments:

Thank you for revising your manuscript according to our editorial requests. While we are satisfied with your revisions overall, I'm afraid we must ask that you further address some of our revision points before we can hand your manuscript to our Production department.

1. Please specify, in the Methods, the source of the bovine vertebrae cancellous bone specimens.
   Reply: We have modified accordingly.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
   Reply: We have done accordingly.